Author's response to reviews

Title: Interprofessional education in primary care for the elderly; a pilot study

Authors:

Barth Oeseburg (b.oeseburg@umcg.nl)
Rudi Hilberts (r.hilberts@umcg.nl)
Truus A Luten (galutenvanoverbeeke@hpst.nl)
Antoinette VM van Etten (a.v.m.van.etten@umcg.nl)
Joris PJ Slaets (j.p.j.slaets@umcg.nl)
Petrie F Roodbol (p.f.roodbol@umcg.nl)

Version: 3 Date: 29 May 2013

Author's response to reviews: see over
Author's response to reviews

Title: Interprofessional education in primary care for the elderly; a pilot study

Authors:

Barth Oeseburg (b.oeseburg@umcg.nl)
Rudi Hilberts (r.hilberts@umcg.nl)
Truus A Luten (galutenvanoverbeeke@hpst.nl)
Antoinette VM van Etten (a.v.m.van.etten@umcg.nl)
Joris PJ Slaets (j.p.j.slaets@umcg.nl)
Petrie F Roodbol (p.f.roodbol@umcg.nl)

Version: 2 Date: 2 April 2013

Author's response to reviews:

The following section is added at the end of the methods paragraph:

Ethical approval

The project received a grant by ZonMW (The National Care for the Elderly Program: 310300003; The Netherlands Organisation for Health Research and Development) and the University Medical Center Groningen (UMCG). The study was presented to the UMCG ethical review board, which did not find further approval necessary.

Also the wordcount Main text, page 1, were adapted from 2984 to 3023.